Author’s response to reviews

Title: Clinical applications and effectiveness of guided implant surgery: A critical review based on randomized controlled trials

Authors:

Marco Colombo (marcocol84@gmail.com)
Carlo Mangano (camangan@gmail.com)
Eitan Mijiritsky (mijiritsky@bezeqint.net)
Mischa Krebs (mischa.krebs@web.de)
Uli Hauschild (hauschild@dentaldesign.biz)
Thomas Fortin (thomas.fortin@univ-lyon1.fr)

Version: 2 Date: 20 Nov 2017

Author’s response to reviews:

Dear Reviewers,

Thank you for your revisions on our work “Clinical applications and effectiveness of guided implant surgery: A critical review based on randomized controlled trials”. Following changes have been made as a result of your records:

• author’s emails have been added;
• conclusions have been rewritten;
• the declaration 'Availability of Data' has been completed with requested information.

We really hope these changes and clarifications are provided can meet with your approval. Thank you very much.

Best regards